Author's response to reviews

Title: Primary hydatid cyst of the gallbladder: A case report

Authors:

Avdyl Krasniqi (dr_krasniqi2001@yahoo.com)
Dalip Limani (DalipLimani@hotmail.com)
Lumturije Gashi-Luc (lumaluci@hotmail.com)
Gazmend Spahija (Gazmend_Spahija@hotmail.com)
Ismail A Dreshaj (ixd4@case.edu)

Version: 3 Date: 15 December 2009

Author's response to reviews: see over
Dear Editor,

Please find attached the revised manuscript number 1880782282307660 entitled ‘Primary hydatid cyst of the gallbladder: A case report’.

Following changes were included:
- Country for each affiliation author
- Ethnic background and nationality of the patient
- A corrected abstract without key words
- Correction of figure numbers and addition of the legend for the figure 3
- Consent section was placed before the competing interests section and
- Finally, some minor revisions were included

I would like to thank you for accepting our manuscript. Please do not hesitate to contact me for any further changes.

Best wishes
Avdyl Krasniqi